Author's response to reviews

Title: Serum tryptase, mast cells positive to tryptase and microvascular density evaluation in early breast cancer patients: possible translational significance

Authors:

Ilaria MARECH (imarech@ibero.it)
Michele AMMENDOLA (Mammedola@gmail.com)
Rosario SACCIO (rsacco@alice.it)
Gennaro S CAPRIOLO (gcapiulo@gmail.com)
Rosa PATRUNO (rpatruno@ibero.it)
Rosangela RUBINI (rrubini@ibero.it)
Maria LUPOSELLA (mluposella@lice.it)
Valeria ZUCCALA (vzuccala@gmail.com)
Eufemia SAVINO (esavino@libero.it)
Cosmo D GADALETA (cgadaleta@libero.it)
Domenico RIBATTI (domenico.ribatti@uniba.it)
Girolamo RANIERI (giroran@tiscali.it)

Version: 6 Date: 2 July 2014

Author's response to reviews: see over
To the Editor-in-Chief

BMC Cancer

Dear Editor,

I send you herewith attached a revised version of the manuscript entitled: “Serum tryptase, mast cells positive to tryptase and microvascular density evaluation in early breast cancer patients: possible translational significance” in which suggestions raised have been answered.

In detail,

Please ensure you include an author contribution for Gennaro S Capriuolo.

I have included an author contribution for Gennaro S Capriuolo.

Please ensure you include spaces between words, especially in the conclusion section of your abstract e.g. "cancerpatients" should be "cancer patients."

I have correctly re-formatted the text.

Please provide a description of your additional files after your current figure legends section.

I have provided a description of the additional files.

Yours sincerely,

Prof. Domenico Ribatti